Author’s response to reviews

Title: Self-efficacy and self-rated oral health among pregnant Aboriginal Australian women

Authors:

Lisa M. Jamieson (lisa.jamieson@adelaide.edu.au)
Eleanor J. Parker (eleanor.parker@adelaide.edu.au)
Kaye F. Roberts-Thomson (kayeroberts-thomson@adelaide.edu.au)
John S. Broughton (john.broughton@otago.ac.nz)
Herenia P. Lawrence (herenia.lawrence@dentistry.utoronto.ca)

Version: 2 Date: 19 December 2013

Author’s response to reviews:

To whom it may concern

We have: (1) amended title page to include author affiliations and contact details; (2) added 'conflict of interest' section; (3) added 'author contribution' section.

Yours sincerely

Lisa M. Jamieson